Author’s response to reviews

Title: In vitro antimicrobial activity of plants used in traditional medicine in Gurage and Silti Zones, south central Ethiopia.

Authors:

Alemtshay Teka (AlemtshayTeka.Sahile@UGent.be)
Johana Rondevaldova (rondevaldova@ftz.czu.cz)
Zemede Asfaw (zasfaw49@yahoo.com)
Sebsebe Demissew (sebseb.demissew@gmail.com)
Patrick Van Damme (Patrick.VanDamme@UGent.be)
Ladislav Kokoska (kokoska@ftz.czu.cz)

Version: 2  Date: 2 May 2015

Author’s response to reviews:

Two pages, including title page, were uploaded but I would like only the second page to be considered as a cover letter.